Comments:

1. In the abstract, please first write the full form of RCT.

Thank you, sir. Full form of RCT is added as advised.

The current meta-analysis includes 8 randomized controlled trials (RCTs) with a total of 1908 T1DM.

2. Line No. 112. Write beta cells instead of b-cells.

Thank you, sir. Beta cells written instead of b-cells added as advised.

It specifically targets the CD3 protein and is believed to reduce the activity of autoreactive T lymphocytes, which are responsible for causing the death of beta-cells.

3. Please add a description to each figure for the understanding of the ordinary reader.

Thank you, sir. Description is added to each figure as advised.

4. The title should be in word form, and the table can be in the form of an image.

Thank you, sir. Title of tables is in the word format as advised.

5. Please follow the standard format for Table 1.

Thank you, sir. Table 1 is formatted as advised.

6. There are 7 Figures and 3 Tables should be improved.

Thank you, sir. Tables and figures are improved as advised.

7. Core tip section. The number of words should be controlled between 50-100 words.

Thank you, sir. Core tip section is provided as advised.

8. Please provide the original figure documents.

Thank you, sir. Original figures are provided and editable form of figure 1 is provided as word file as well along with JPEG image.

9. Uniform presentation should be used for figures showing the same or similar contents

Thank you, sir. Uniform presentation is used for figures as advised.